Author’s response to reviews

Title: Effect of a peer-educational intervention on provider knowledge and reported performance in family planning services: a cluster randomized trial

Authors:

Sakieh Mohammad-Alizadeh Charandabi (Sakineh.mohammad.alizadeh@ki.se)
Rezagoli Vahidi (vahidireza@yahoo.com)
Lena Marions (lena.marions@karolinska.se)
Rolf Wahlström (rolf.wahlstrom@ki.se)

Version: 4 Date: 29 December 2009

Author’s response to reviews: see over
Dear Nina Titmus,

Thanks a lot for the valuable comments to improve our manuscript. I tried to perform all your comments as following:

1. I had forgotten to write about informed consent from the participants. So, I added "Written informed consent was obtained from all participants" to the end of paragraph 3, page 7.

2. I tried to perform necessary changes to conform to the journal style. Some of the changes are:
   - Adding e-mail of all authors to the title page.
   - Excluding sub-title of "Objective" from Abstract, Changing title of "Introduction" to "Background", Adding title of "Conclusion", ...
   - Changing style of writing the references.

3. I sent the figure and Annex in separate files.

Sakineh Mohammad-Alizadeh

Corresponding author of the manuscript